Author’s response to reviews

Title: Prevalence and severity of menopause symptoms among perimenopausal and postmenopausal women aged 30-49 years in Gulele sub-city of Addis Ababa, Ethiopia

Authors:

Engida Yisma (engiday@gmail.com)

Natnael Eshetu (natnaelesetu87@gmail.com)

Stephanie Ly (stephaniely@ucla.edu)

Berhanu Dessalegn (berhanudessa@yahoo.com)

Version: 3 Date: 31 Aug 2017

Author’s response to reviews:

Point-by-point response to editor’s comments on the manuscript entitled “Prevalence and severity of menopause symptoms among perimenopausal and postmenopausal women aged 30-49 years in Gulele sub-city of Addis Ababa, Ethiopia” (MS: BMWH-D-16-00086R2).

Authors:

Engida Yisma (engiday@gmail.com)

Natnael Eshetu (natnaelesetu87@gmail.com)

Stephanie Ly (stephaniely@ucla.edu)

Berhanu Dessalegn (berhanudessa@yahoo.com)

We are very grateful for the valuable comments that the editor have provided us. We have taken due consideration of the comments provided and made all necessary revisions to the manuscript. We have provided a point-by-point response to the comments as follows.

Response to Editor’s Comments

COMMENT: Please format your Availability of data and materials statement according to the guidelines

AUTHORS’ RESPONSE: We have formatted “Availability of data and materials section” as per the guidelines in the revised manuscript.
COMMENT: In order for your questionnaires to be published as supplementary information, you must include in your manuscript a list of additional files.

AUTHORS’ RESPONSE: Thank you so much! We have included a list of additional files in the revised manuscript.

COMMENT: Please submit a clean, final version of your manuscript without any highlighting and with track changes turned off.

AUTHORS’ RESPONSE: We have submitted the manuscript accordingly.